Author’s response to reviews

Title: Euthanasia requests in Dementia Cases; what are experiences and needs of Dutch physicians? A qualitative interview Study

Authors:

Jaap Schuurmans (jschuur@knmg.nl)
Romy Bouwmeester (romy.bouwmeester@radboudumc.nl)
Lamar Crombach (lamar.crombach@live.nl)
Tessa van Rijssel (t.vanrijssel@student.ru.nl)
Lizzy Wingens (lizzy.w@hotmail.com)
Kristina Georgieva (k.georgieva@student.ru.nl)
Nadine O'Shea (N.O'Shea@student.ru.nl)
Stephanie Vos (stephanie.vos@radboudumc.nl)
Bram Tilburgs (bram.tilburgs@radboudumc.nl)
Yvonne Engels (yvonne.engels@radboudumc.nl)

Version: 2 Date: 26 Aug 2019

Author’s response to reviews:

Dear Oliver Cocks and Janelle Coore,

Thank you for assessing our revised manuscript and its potential to be accepted in your journal.

Below, you will find our reaction to the editors comments:

1. We changed the Availability of Data and Materials section accordingly. (page 14). Data and materials are available from the corresponding author on reasonable request.

2. LQ was a typing error, this should be LW.
3. We adapted the individual contributions of all authors' section. (page 14): JS, RB, LC, Tvr, LW, KG, NO'S, BT and YE contributed to the design of the study. ..... 

4. All authors have read and approved the manuscript, which we also added to the authors contribution section on page 14.

5. We uploaded the adapted, clean version.

As the position of last author Yvonne Engels has changed to full professor in August, we adapted this too.

We also checked the Editorial Policies.

On behalf of all authors,

Best wishes, Jaap Schuurmans